Author's response to reviews

Title: Arthroscopic Decompression and Notchplasty for Longstanding Anterior Cruciate Ligament Impingement in a Patient with Multiple Epiphyseal Dysplasia (MED): A Case Report

Authors:

Ravi K Trehan SpR (trehanravi@hotmail.com)
N Dabbas ST II (ndabbas@gmail.com)
D Allwood SHO (dominique23@btinternet.com)
M Agarwal SpR (bonydoc@hotmail.com)
C Kinmont Consultant (christian.kinmont@mayday.nhs.uk)

Version: 5 Date: 2 May 2008

Author's response to reviews: see over
Date: 2nd May 2008

Editorial Board
Journal of Medical case Report

Dear Sir / Ma’am

RE: Arthroscopic Decompression and Notchplasty for Longstanding Anterior Cruciate Ligament Impingement in a Patient with Multiple Epiphyseal Dysplasia (MED): A Case Report

I have done following changes in our manuscript as advised by the deputy editor:

1- References were already put for Oxford knee score and SF-36 in last revision dated 12/04/08.
2- Last paragraph of discussion is dropped.

Thanking you

Sincerely yours

Mr Ravi Trehan
22a, York Road
Cheam
Surrey
SM2 6HH

trehanravi@hotmail.com